Author's response to reviews

Title: Excellent outcomes among HIV+ children on ART, but unacceptably high pre-ART mortality and losses to follow-up: a cohort study from Cambodia

Authors:
Marie-Eve Raguenaud (eve_raguenaud@hotmail.com)
Petros Isaakidis (petrosisaakidis@yahoo.com)
Rony Zachariah (rony.zacharia@brussels.msf.org)
Vantha Te (cdc_takeo@camintel.com)
Seithabot Soeung (varun@angkorhospital.org)
Kazumi Akao (kazumiakao0520@hotmail.com)
Varun Kumar (varun@angkorhospital.org)

Version: 5 Date: 28 July 2009

Author's response to reviews: see over
Dear Editor,

Thank you for this last review.

We have now revised the paper in line with the final remark from the reviewer.

Please find the following for your kind consideration:

1) The response to the comment of the reviewer
2) A new revised version of the manuscript.

We now hope that these changes will meet with your favourable consideration.

Yours Sincerely,

Marie-Eve Raguenaud
Author for correspondence
Cover letter: reply to reviewers' comments

MS: 2000759564248192
Title: Excellent outcomes among HIV+ children on ART, but unacceptably high pre-ART mortality and losses to follow-up: a cohort study from Cambodia

Date: July 26, 2009

Reply to reviewer Margaret May:

Minor revision:

On p.8 (mortality section) we have added the sentence suggested to explain the contents of Table 2.